Author's response to reviews

Title: An Empirical Study to Determine Factors That Motivate and Limit The Implementation Of ICT In Healthcare Environments

Authors:

Raj Gururajan (<u>gururaja@usq.edu.au</u>) Abdul Hafeez-Baig (<u>abdulhb@usq.edu.au</u>)

Version: 3 Date: 5 September 2013

Author's response to reviews:

We have added a statement to confirm that institutional ethics approval was sought and obtained for data collection. This statement was added i the methods section, above the Multiple Case Studies Section.